Author’s response to reviews

Title: "Provoking conversations": case studies of organizations where Option GridTM decision aids have become 'normalized'

Authors:

Peter Scalia (peter.scalia@dartmouth.edu)
Glyn Elwyn (glynelwyn@gmail.com)
Marie-Anne Durand (durand.marieanne@gmail.com)

Version: 1 Date: 01 May 2017

Author’s response to reviews:

Dear Reviewer,

Thank you very much for your feedback. We apologize for not including the required information in the original manuscript. We added two sentences to the 'Ethics approval and consent to participate' section of the 'Declarations' page.

We added: "The study received ethical approval from the Dartmouth College Committee for the Protection of Human Subjects (STUDY00028711). All individuals voluntarily participated by reading a 'consent to take part in research’ form and providing written consent to participate."

We, therefore, stated that ethics approval was obtained and provided the name and reference number of the ethics committee. We also provided details on consent that was obtained from each participant.

We greatly appreciate your feedback.

Thank you

Peter Scalia MSc.